Author's response to reviews

Title: The process of developing and implementing a telephone-based peer support program for postpartum depression: lessons learned from several randomized controlled trials

Authors:

   Cindy-Lee Dennis (cindylee.dennis@utoronto.ca)

Version: 2 Date: 25 September 2013

Author's response to reviews: see over
September 25, 2013

Trials
c/o BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom
Email: editorial@trialsjournal.com

Dear Sir/Madam,

Re: Revised Manuscript Submission

Please find enclosed the manuscript: “The process of developing and implementing a telephone-based peer support program for postpartum depression: lessons learned from several randomized controlled trials,” submitted as a research article to Trials for consideration of publication.

Changes in the manuscript based on the Senior Editorial Assistant’s instructions are outlined as follows:

1. I have included a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval, and a statement explaining that informed consent was obtained from each participant.

Thank you for allowing me to make this change and for considering my manuscript for publication.

Sincerely,

Dr. Cindy-Lee Dennis
University of Toronto
Lawrence S. Bloomberg Faculty of Nursing
155 College St
Toronto, ON M5T 1P8
Tel: (416) 946-8608
Fax: (416) 978-8222
Email: cindylee.dennis@utoronto.ca